Author's response to reviews

Title: Quality improvement collaboratives and the wisdom of crowds: spread explained by perceived success at group level

Authors:

   Michel L.A. Dückers (m.duckers@impact.arq.org)
   Peter P. Groenewegen (p.groenewegen@nivel.nl)
   Cordula Wagner (c.wagner@nivel.nl)

Version: 3 Date: 26 June 2014

Author's response to reviews:

Approved the copy-edited version.

Modifications:

1. Page number is added between parenthesis after quote Ovretveit et al. 2002.
2. In Table 2 (final page), the text in parenthesis is clarified:
   * Between parentheses, total number of project team leaders providing a success score
   ** Between parentheses, total number of programme coordinators providing dissemination data